Author's response to reviews

Title: Observed Communication Skills: How Do They Relate To The Consultation Content? A nation-wide study of graduate medical students seeing a standardized patient for a first-time consultation in a general practice setting.

Authors:

Tore Gude (tore.gude@medisin.uio.no)
Per Vaglum (p.j.w.vaglum@medisin.uio.no)
Tor Anvik (tor.anvik@ism.uit.no)
Anders Baerheim (anders.barheim@isf.uib.no)
Hilde Eide (Hilde.Eide@su.hio.no)
Ole Fasmer (ole.fasmer@psyk.uib.no)
Peter Graugaard (p.k.gravgaard@medisin.uio.no)
Hilde Grimstad (hilde.grimstad@medisin.ntnu.no)
Per Hjortdahl (per.hjortdahl@medisin.uio.no)
Are Holen (are.holen@medisin.ntnu.no)
Tone Nordoy (tone.nordoy@unn.no)
Helge Skirbekk (helge.skirbekk@medisin.uio.no)
Arnstein Finset (arnstein.finset@medisin.uio.no)

Version: 4 Date: 27 September 2007

Author's response to reviews: see over
Dear Lolu da-Silva

Please find enclosed my revised paper according to the reviewers’ comments:

LuAnn Wilkerson: - The title on Table 4 is adjusted by eliminating the curricular type
- References for Aspegren (p. 9) and Pfeiffer (p. 13) are included.

Nobutaro BAN - No special comments except for the language, as also the first reviewer
did comment.

When I check the References in the special window, there is in Ref, nr, 1 a lack of names
written properly for Silverman, J and Draper, J, in both my former and the revised manuscript
the names are written correctly.

The manuscript has been reviewed by a native English-speaking psychologist.

On behalf of my co-authors and myself, I thank for the positive comments from the reviewers
and the acceptance of this paper for publication in BMC Medical Education.